Dear Editor and Reviewers,

Thank you for your letter and for the reviewers’ comments concerning our manuscript. We greatly valued all the comments, which were very helpful for us to revise the paper and to improve its quality. Based on your comments, we have made a careful revision throughout the manuscript. To the best of our abilities, we modified the manuscript to improve its understandability and scientific quality.

PEER-REVIEW REPORT

Name of journal: World Journal of Clinical Cases

Manuscript NO: 70169

Title: Fluoroscopic removal of fractured, retained, embedded Z self-expanding metal stents using a guidewire lasso technique-a case report

Reviewer’s code: 05151858

Position: Peer Reviewer

Academic degree: PhD

Professional title: Doctor

Reviewer’s Country/Territory: Turkey
SPECIFIC COMMENTS TO AUTHORS
The author described a rare case. A successful operation was performed. Post-treatment radiography and CT images are not very successful, but in such cases, it is not possible to go back and take them again. If there is no other problem, I find it publishable. Kind regards.

Answer: The manuscript was revised by native English speaker, and the figures was provided in PPT, thanks.